Author’s response to reviews

Title: Importance of social capital at the workplace for return to work among women with a history of long-term sick leave: a cohort study

Authors:

Ingela Rydström (ingela.rydstrom@hb.se; linda.ahlstrom@hb.se)

Ann-Charlotte Dalheim Englund (lotta.englund@hb.se)

Lotta Dellve (lotta.dellve@socav.gu.se)

Linda Ahlstrom (linda.ahlstrom@gu.se)

Version: 2 Date: 09 Jun 2017

Author’s response to reviews:

Dear Editor and Reviewer,

Thank you for your review of our manuscript “Importance of social capital at the workplace for return to work among women with a history of long-term sick leave: a cohort study”. We have taken your comments into consideration (NURS-D-17-00043R1) and added a Declarations section before the references and done some changes in the tables. We have chosen to highlight our additions and changes in the text and hope you find this choice relevant. We also want to add that we are sad that parts of the Declaration section disappeared when we made the changes that were requested during previous review process.

Yours sincerely

Ingela Rydström